Author’s response to reviews

Title: Consumption of whole grains and legumes modulates the genetic effect of the APOA5 -1131C variant on changes in triglyceride and apolipoprotein A-V concentrations in patients with impaired fasting glucose or newly diagnosed type 2 diabetes: an open label randomized controlled intervention trial

Authors:

Ryungwoo Kang (lovesky@hanmail.net)
Minjoo Kim (minjookim@yonsei.ac.kr)
Jey Sook Chae (jeysookchae@yonsei.ac.kr)
Sang-Hyun Lee (keroboom@nate.com)
Jong Ho Lee (jhleeb@yonsei.ac.kr)

Version: 3
Date: 8 January 2014

Author’s response to reviews: see over
Dear Editor,

We are happy to submit the revise version of manuscript (7032888771014570) to Trials. We sincerely appreciate the time spent in reviewing this manuscript and reviewers’ advices to improve it. Please, see the attached files (revised manuscript, point-by point answers to reviewers’ queries and comments). We marked the correction in red. We hope that you find them satisfactory. The principle author and all co-authors have read and approved the submission of the revised version of manuscript; the manuscript has not been published and is not being considered for publication elsewhere, in whole or in part, in any language. And there will not be any potential conflict of interest.

**MS : 7032888771014570 (1st revision)**

Consumption of whole grains and legumes modulates the genetic effect of the *APOA5*-1131C variant on changes in triglyceride and apolipoprotein A-V concentrations in patients with impaired fasting glucose or newly diagnosed type 2 diabetes: an open label randomized controlled intervention trial

**Corresponding author & Address:** Jong Ho Lee
Department of Food & Nutrition, College of Human Ecology, Yonsei University
50 Yonsei-ro, Seodaemun-gu, Seoul, 120-749, Korea
Fax: +82-2-364-9605; Tel: +82-2-2123-3122; E-mail: jhleeb@yonsei.ac.kr

Thank you very much for your time.

Sincerely yours,

Jong Ho Lee
Professor,
Department of Food & Nutrition,
College of Human Ecology, Yonsei University
50 Yonsei-ro, Seodaemun-gu, Seoul, 120-749, Korea
Fax: +82-2-364-9605; Tel: +82-2-2123-3122;
E-mail: jhleeb@yonsei.ac.kr